Author's response to reviews

Title: Assessment of the effects of a static magnetic field on bone healing process in rat femurs

Authors:

- Edela EP Puricelli (epuricelli@uol.com.br)
- Lucienne LU Ulbrich (lmulbrich@yahoo.com)
- Deise DP Ponzoni (deponzoni@yahoo.com)
- Joao Julio JJCF Cunha (jjulio.voy@terra.com.br)

Version: 2 Date: 22 May 2006

Author's response to reviews:

Dear Sir

We would like to thank for your answer and for the review of the referees. We are sending you the manuscript and figures, modified according to the suggestions. Please let us know if you received them, and if any other modification is needed.

The news figures sending in other e-mail in sequential of this. We would like to ask you to send any comment or question to the main author, Dr Edela Puricelli, puricelli@uol.com.br.

Yours sincerely,

Prof. Dr. Edela Puricelli
President-ALACIBU